Author’s response to reviews

Title: Improving KT tools and products: Development and evaluation of a framework for creating optimized, Knowledge-activated Tools (KaT)

Authors:

Monika Kastner (monika.kastner@utoronto.ca; monika.kastner@nygh.on.ca)
Julie Makarski (julie.makarski@nygh.on.ca)
Leigh Hayden (leigh.hayden@sheridancollage.ca)
Yonda Lai (yonda.lai@gmail.com)
Joyce Chan (joyce.chan.77@gmail.com)
Victoria Treister (vtreister@gmail.com)
Kegan Harris (kegan.harris55@gmail.com)
Sarah Munce (sarah.munce@uhn.ca)
Jayna Holroyd-Leduc (jayna.holroyd-leduc@albertahealthservices.ca)
Ian Graham (igraham@ohri.ca)
Sharon Straus (sharon.straus@utoronto.ca)

Version: 2 Date: 16 Mar 2020

Author’s response to reviews:

Thank you very much for your comments and suggestions. We made the following edits:

1. We made the change to the abstract, "Conclusions” section (Page 3) to reflect the limitation that our survey participants were mostly Canadian: "Our findings suggest that mostly Canadian KT experts and knowledge users perceived the KaT framework and the future development of an online, interactive version to be important and needed."

2. We indicated where Appendix E is referred to in the manuscript.

Thanks so much,

Monika